Author's response to reviews

Title: Loss of Endocan tumorigenic properties after alternative splicing of exon 2

Authors:

Florence Depontieu (fdepont2@jhmi.edu)
Bogdan D Grigoriu (b_grigoriu@hotmail.com)
Arnaud Scherpereel (a-scherpereel@chru-lille.fr)
Estelle Adam (estelle_adam@voila.fr)
Maryse Delehedde (mdelehedde@yahoo.fr)
Philippe Gosset (philippe.gosset@pasteur-lille.fr)
Philippe Lassalle (philippe.lassalle@pasteur-lille.fr)

Version: 2 Date: 24 October 2007

Author's response to reviews:

Dear Editor,

In the name of all authors we thank you and the two reviewers for evaluating our manuscript entitled “Loss of Endocan tumorigenic properties after alternative splicing of exon 2”.

Please find along with this letter the revised form of the manuscript along with the point by point response to reviewer’s comments.

Sincerely Yours,
Dr. Bogdan Dragos GRIGORIU